Author's response to reviews

Title: Impact of oral melatonin on adult critically ill patients with ICU sleep deprivation: a study protocol and statistical analysis plan for a randomized controlled trial

Authors:

Huawei Huang (vivian198543@163.com)
Li Jiang (jiangli@sina.com)
Bo Zhu (zhubo123123@sina.com)
Jiajia Cheng (343155552@qq.com)
Xiuming Xi (xxm_fxyy@sina.cn)

Version: 2
Date: 19 March 2014

Author's response to reviews:

Dear Editor:

Thank you for your letter. I am very sorry that the trial registration number we provided is incorrect from carelessness. Our trial was successfully registered in the Chinese Clinical Trial Registry, and its correct registration number is ClinicalTrials (NCT): ChiCTR-TRC-14004319. We are terribly sorry for the inconvenience brought to you.

Best wishes,

Xiuming Xi